Author's response to reviews

Title: Polyunsaturated fatty acid intake and prevalence of eczema and rhinoconjunctivitis in Japanese children: The Ryukyus Child Health Study

Authors:

Yoshihiro Miyake (miyake-y@fukuoka-u.ac.jp)
Keiko Tanaka (k-tanaka@fukuoka-u.ac.jp)
Satoshi Sasaki (stssasak@m.u-tokyo.ac.jp)
Masashi Arakawa (h069475@ll.u-ryukyu.ac.jp)

Version: 2 Date: 24 December 2010

Author's response to reviews:

We added the following to the last of the Study Population section in Methods: Written informed consent was obtained from all subjects; in this survey, a signature of the children/adolescent in the questionnaires was considered informed consent made by both the children/adolescent and the parent(s)/caregiver(s).